Author's response to reviews

Title: The burden of malaria to the poorest household is unjust but redeemable: Inequities in incidence, morbidity and expenditures on prevention and treatment in southeast Nigeria

Authors:

Nkolika P Uguru (nk_uguru@yahoo.com)
Obinna E Onwujekwe (onwujekwe@yahoo.co.uk)
Benjamin S Uzochukwu (bscuzochukwu@yahoo.com)
Godwin C Igiliegbe (igiliegbechuks@yahoo.com)
Soludo B Eze (soludo2001@yahoo.co.uk)

Version: 2 Date: 24 September 2008

Author's response to reviews:

The revision being made is to the methods section. I want to add that approval of this research was received from the ethical clearance committee of the College of Medicine University of Nigeria Nsukka Enugu Campus Enugu State Nigeria.